Author’s response to reviews

Title: Reduced transient receptor potential vanilloid 2 expression in alveolar macrophages causes COPD in mice through impaired phagocytic activity

Authors:

Hiroaki Masubuchi (hmasubuc@gunma-u.ac.jp)
Manabu Ueno (manabun0825@gmail.com)
Toshitaka Maeno (mutoyu03@gunma-u.ac.jp)
Koichi Yamaguchi (ckpnt341@yahoo.co.jp)
Kenichiro Hara (pourta5311@yahoo.co.jp)
Hiroaki Sunaga (hsunaga@gunma-u.ac.jp)
Hiroki Matsui (hmatsui@gunma-u.ac.jp)
Masahiro Nagasawa (88zzhospital@gmail.com)
Itaru Kojima (ikojima@gunma-u.ac.jp)
Yuko Iwata (yukoiwat@ncvc.go.jp)
Shigeo Wakabayashi (pha047@osaka-med.ac.jp)
Masahiko Kurabayashi (mkuraba@gunma-u.ac.jp)

Version: 2 Date: 03 Jan 2019

Author’s response to reviews:

We would like to thank the reviewer #1 for providing us with valuable comments that has allowed us to improve the quality of our manuscript.

We added the sentence or section about 1. Euthanasia (page 9, line 1), 2. Finding (page 18, line 23), 3. Author’s contributions (page 19, line 4) and 4. Additional files (page 25, line 1) according to your suggestion.